Author’s response to reviews

Title: Predictors of the late onset of cannabis and other drug use in male young adults

Authors:

Severin Haug (severin.haug@isgf.uzh.ch)
Carla López Núñez (carlalopezn@gmail.com)
Julia Becker (julia.becker@isgf.uzh.ch)
Gerhard Gmel (ggmel@addictionsuisse.ch)
Michael P Schaub (michael.schaub@isgf.uzh.ch)

Version: 2 Date: 3 March 2014

Author’s response to reviews:

Dear Editors

Within the revised manuscript we added the following sections:

Conclusions
List of abbreviations used
Competing interests
Authors’ contributions
Acknowledgements

Kind Regards

Severin Haug